Author's response to reviews

Title: Standardization of examination posture used for spinal curvature assessment by means of the DTP-3

Authors:

Jakub Krejci (jakub.krejci@upol.cz)
Jiri Gallo (jiri.gallo@volny.cz)
Petr Stepanik (petr.stepanik@upol.cz)
Jiri Salinger (jsalinger@volny.cz)

Version: 2 Date: 4 January 2012

Author's response to reviews: see over
Dear Editor,

please find attached the revised manuscript (MS) newly entitled "Optimization of examination posture used for spinal curvature assessment" by Krejci, Gallo et al.

Change of the title and other changes in the manuscript that were induced by the reviewers’ comments are marked red. In addition, attached is list of “Responses” on reviewers’ criticism.

We are the authors responsible for the integrity of the study. We and our co-authors, who all read and approved the MS, would like to re-submit it for consideration for publication to the "Scoliosis".

No part of the research presented in the MS has been submitted for elsewhere nor is being submitted elsewhere simultaneously.

The work has not been supported from commercial sources and we are not aware of any potential conflict of interests.

When preparing the manuscript for potential publishing in the "Scoliosis", all available instructions were respected.

Thank you for an opportunity to re-submit our work and for letting us know your editorial decision in due course.

Yours sincerely,

Jakub Krejci, PhD
Palacky University Olomouc, Czech Republic

Jiri Gallo, MD, PhD, corresponding author
Palacky University Olomouc, Czech Republic